THE ECONOMISTS AND THE PRESS IN ITALY FROM THE END OF THE NINETEENTH CENTURY UNTIL FASCISM: THE CASE OF LUIGI EINAUDI

BY GIOVANNI PAVANELLI

Luigi Einaudi was an authoritative Italian economist and a leading representative of economic and political liberalism in Europe. After the Second World War, he became governor of the Bank of Italy and president of the Italian Republic. This paper analyzes his role as opinion maker from the end of the nineteenth century until the 1920s, when he was a leading columnist at La Stampa and the Corriere della Sera, the most influential newspapers in Italy at that time. It focuses on the scope and limits of Einaudi's efforts to broaden consensus in Italian public opinion on the principles of economic liberalism and free competition. To this end, it investigates Einaudi's journalistic style, his views on the role of the newspapers, and his large following among the public. Further sections analyze the main issues tackled by Einaudi in his articles in the Corriere and the systematic work of propaganda he enacted during World War I to convince the Italian households to reduce consumption and to finance the military expenditure. A final section deals with the "reconstruction program" devised by Einaudi in the early 1920s with the aim of restoring price stability and fiscal restraint, his efforts in the Corriere to propagate this program, and his forced retreat from journalism after the beginning of the fascist regime in 1924-25.

Giovanni Pavanelli: Department of Economics, Mathematics and Statistics ESOMAS, University of Turin. E-mail: giovanni.pavanelli@unito.it. This research draws on a wider project on economics and public opinion in Italy during the "liberal age" whose results are published in Augello, Guidi, and Pavanelli (2016). Previous drafts of this paper were presented at a seminar held at the VSE (University of Economics), Prague, and at the fourth annual MRC Conference, UNWE Sofia, 2018. I wish to thank Massimo M. Augello, Enrico Colombatto, Pedro G. Duarte, Marco E. L. Guidi, David Lipka, Nikolay Nenovsky, and two anonymous referees for their comments. The usual disclaimer applies.

ISSN 1053-8372 print; ISSN 1469-9656 online/23/04000576-602 © The Author(s), 2023. Published by Cambridge University Press on behalf of the History of Economics Society. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0), which permits unrestricted re-use, distribution and reproduction, provided the original article is properly cited. doi:10.1017/S1053837222000438

I. INTRODUCTION

A distinguishing feature of the Italian economists' work from the second half of the nineteenth century to the present day appears to be a close interaction between theoretical analysis and a thorough and passionate reflection on the complex issues stemming from Italy's institutional and economic transformation.

Recent research has allowed us to grasp at least partially this multifaceted activity, focusing particularly on the role played by the economists as legislators and as advisors on policy matters, and on their work as columnists and opinion makers in the daily press (Augello and Guidi 2001, 2003, 2005; Augello, Guidi, and Pavanelli 2016). As far as this latter issue is concerned, there is indeed a large consensus on the fact that newspapers played a key role in transmitting ideas and were a powerful instrument in shaping public opinion in Italy during the "liberal age" (Castronovo 1976; Castronovo and Tranfaglia 1979; Murialdi 1997). Public opinion, in turn, became increasingly influential starting from the 1880s and 1890s, as a consequence of a surge in literacy and a gradual widening of voting rights.

On this point, it is crucial to consider that in the early Risorgimento,² newspapers—albeit often influential—reached only limited sectors of the population, namely the élite taking part in political activity. It was only in the last two decades of the nineteenth century that some newspapers started being published and distributed at a national level, reaching a wider audience and becoming authoritative: among others, *La Stampa* printed in Turin; *Il Sole, Corriere della Sera*, and *Il Secolo*, Milan; *La Tribuna*, *Il Giornale d'Italia*, and *Il Tempo*, Rome; and *Il Mattino*, Naples. A key role was also played by newspapers connected to political parties and social movements, such as the *Avanti!* and *Popolo d'Italia*.³ Starting from this period and until the forced suspension of the freedom of the press imposed by the fascist regime, newspapers were one of the most important instruments in interpreting the needs and influencing the choices of the public, in transmitting ideas and programs, in advocating economic and social policy measures, and in explaining and criticizing those adopted by the central government and by local administrators.⁴

In this framework, the Italian economists played a crucial and, until recently, largely neglected role as columnists and opinion makers. Let us mention, amongst others, Maffeo Pantaleoni, Vilfredo Pareto, Luigi Luzzatti, Antonio De Viti de Marco, Luigi Einaudi, Attilio Cabiati, and Gino Borgatta.⁵ This paper focuses on Luigi Einaudi's

¹ The daily press played indeed a key role in disseminating ideas and policy proposals in most industrialized countries in Europe and North America: "Newspapers," writes Wayne Parsons, "could be said to lie at the very core of the capitalist process ... and consequently have a key role to play in the way in which 'economic culture' is communicated and its myths and discourse sustained and propagated" (Parsons 1989, p. 2).

² The Risorgimento is the period of Italian history, beginning in 1815, in which the process of political unification of the country took place. The proclamation of the kingdom of Italy and the completion of the first phase of the unification occurred in 1861, but usually the conclusion of the Risorgimento in set in 1871, the year of the transfer of the capital to Rome.

³ Avanti!, published since 1896, was the official newspaper of the Italian socialist party. *Popolo d'Italia*, founded in 1914, became in the early 1920s the official newspaper of the Italian fascist party.

⁴ On the of the role of the newspapers in Italy in this context, see Forgacs (1990); Brizzi (2012).

⁵ The work of these economists as opinion makers is analyzed in Augello, Guidi, and Pavanelli (2016). For a broad overview of the Italian economic thought, see Faucci (2014).

contribution, which is indeed striking: from 1896 till 1925, when he was forced to give up writing as a columnist in consequence of increasing pressure by the fascist regime, Einaudi published about 400 articles in *La Stampa* and about 1,700 in the *Corriere della Sera*. In these pieces, partially reprinted in an eight-volume edition entitled *Cronache economiche e politiche di un trentennio* (Economic and political chronicles of thirty years), he critically analyzed the most relevant issues facing the Italian economy and society of his time and campaigned in favor of free competition, monetary stability, and fiscal restraint (Pavanelli 2012, 2016).

Luigi Einaudi (1874–1961) was undoubtedly one of the most influential Italian economists of the first half of the twentieth century and, at the same time, a leading representative of liberalism in Europe. Professor of public finance at the University of Turin from 1902, he had an impressive intellectual activity, which spanned over nearly sixty years. He was a gifted and prolific scholar, and published numerous essays and monographs in the fields of public economics, monetary and fiscal policy, economic history, and history of economic thought. He also wrote original contributions on political issues. Similarly to other Italian intellectuals, at the beginning of the 1920s, he expressed conditional support to the fascist movement, seen as a way to restore domestic order after the dramatic political turmoil of the post-World War I period. After 1924, however, he—like many others—became a resolute opponent of the fascist regime. Member of the Italian Senate since 1919, in 1944 he was appointed by the new anti-fascist government as head of the Bank of Italy and three years later was called by Alcide De Gasperi, then leader of a centrist coalition, to be part of his cabinet as the minister of the budget. The following year, he was elected by Parliament as president of the newly established Republic of Italy.

This paper aims to analyze Einaudi's role as opinion maker from the end of the nineteenth century until the early 1920s, when he was the leading columnist on economic issues in *La Stampa* and the *Corriere della Sera*. It focuses in particular on the scope and limits of his efforts to broaden consensus in Italian public opinion on the principles of economic liberalism, government's budget balancing, and monetary stability. Although secondary literature on Einaudi is extensive,⁶ several aspects of his work as opinion maker and columnist remain to be properly investigated. In particular, I explore the connections between his activities as a newspaper columnist and his role as a public intellectual committed to intervene on the fundamental policy debates of his time on the basis of a clearly defined philosophical system, that of liberalism. Insufficient attention has also been paid so far to the constraints, next to the potentialities, Einaudi had to deal with in conveying his ideas through modern newspapers such as *La Stampa* and *Corriere della Sera*, structured as profit-oriented enterprises, attentive to the tastes and expectations of their readers and characterized by specific editorial policies. Furthermore, while literature on Einaudi's work as columnist

⁶ On Einaudi's intellectual biography, see Faucci (1986, 1993); Caffè (1987); Cafagna (2004). On his work on public finance, monetary economics, and economic policy, see in particular: Leoni (1964); Forte (1982, 2009); Monti (1990); Luca Einaudi, Faucci, and Marchionatti (2006); Porta (2010); Baffigi (2010); Bini (2010); Forte and Marchionatti (2012); Mattei (2017); Nerozzi (2021). On his work on economic history and history of economic thought, see Romeo (1974). On his contribution to political science and liberalism, see Bobbio (1974); Salvadori (2005); Giordano (2006); Silvestri (2008); Heritier and Silvestri (2012); Masini (2012); da Empoli, Malandrino, and Zanone (2014).

is based mainly on the articles republished by Einaudi himself in *Cronache*, this paper also takes into consideration numerous pieces he wrote for the *Corriere* but not included in the above-mentioned collection and which appear relevant for an assessment of his contribution.

With these premises, this work is organized in the following way: section II contains a short intellectual biography of Einaudi; section III examines Einaudi's journalistic style, his following among the public and the political élite, and his views concerning the role of the newspapers and of the economists as columnists in shaping public opinion; section IV analyzes the main issues tackled by Einaudi in his articles in the daily press before World War I; section V focuses on the work of persuasion and propaganda enacted by him during the war in order to convince the Italian households to reduce consumption and to buy Treasury bills; section VI analyzes the "reconstruction program" devised by Einaudi after World War I and mainly based on economic liberalism, his efforts to propagate this program in the *Corriere*, and his forced retreat from journalism, at least during the interwar period, following the beginning of the fascist regime in 1924–25. Overall, this paper aims to bring new insights into the way the economic discourse is propagated in modern societies and into the role of the economists as opinion makers.

II. EINAUDI'S SHORT INTELLECTUAL BIOGRAPHY

Luigi Einaudi was born in 1874 in Carrù, an agricultural town in Piedmont, to a family belonging to the small rural bourgeoisie. In 1891 he enrolled at the University of Turin to take a degree in law, which included the study of political economy and public finance. While still a student, he started to attend regularly the Laboratorio di economia politica created in 1893 at the Faculty of Law. The Laboratorio, which in a few years would become one of the most important centers in economic research in Italy, had the mission to promote empirical analysis, according to the positivist approach prevailing at that time (Faucci 1995, 2004; Becchio 2004). In the meantime, in 1896, he started to work as a journalist at the newspaper *La Stampa*, published in Turin.

The research work Einaudi carried out in this period at the Laboratorio was the basis of essays and monographs (Einaudi 1900a, 1900b, 1902), which allowed him to obtain in 1902 the chair of public finance at the University of Turin, a position he held until the late 1950s. He was also in charge of the teaching of economics at the University L. Bocconi of Milan until 1926, when he was forced to resign as a consequence of his anti-fascist positions.

The late 1890s and the early years of the new century were crucial for Einaudi's intellectual formation: during this period he built his theoretical foundations as an economist on the works of the classical economists, in particular David Ricardo's monetary writings and John Stuart Mill's *Principles*, and of neoclassical authors such

⁷ During the nineteenth and early twentieth centuries, the Italian universities did not offer degree programs in economics. The study of commercial and economic disciplines was possible within a few *Scuole superiori di commercio* (high schools of commerce) that depended on the Ministry of Agriculture, Industry and Commerce. The degrees conferred by these institutions, however, were considered less prestigious than standard university degrees. In these years, therefore, most of the Italian economists graduated in law.

as Alfred Marshall and, to lesser extent, Vilfredo Pareto. He also drew inspiration from Irving Fisher, particularly for the concept of double taxation of savings (Forte and Marchionatti 2012). In this period, furthermore, he set aside an initial sympathy for the socialist movement and developed a firm commitment to the principles of economic liberalism. On this point, it is crucial to underline that Einaudi's defense of competition and market economy was mainly based on strong ethical considerations rather than on criteria of efficiency: only in a decentralized economy, in which people did not depend on the government for their survival, he maintained, could individuals really be free (Einaudi [1949] 1964, pp. 321–325). To these principles he remained faithful throughout his entire life. In the 1940s, in the face of the social disruptions caused by the Great Depression and the war, he became, however, also a supporter of measures aimed at reducing social inequality, on the basis of the principle of equality at the starting point.

From the first decade of the twentieth century, he became chief editor of *La Riforma Sociale*, an academic journal published in Turin, which under his direction focused on policy issues and became the point of reference of a group of committed liberal economists (the so-called Turin school of economics; cf. Marchionatti, Cassata, Becchio, and Mornati 2013). Besides being an active columnist for the *Corriere della Sera*, during the same period Einaudi became a correspondent for *The Economist*. This concurrent work, started in 1908, was at first sporadic but became regular in the early 1920s by invitation of the editor of the British magazine and continued during the fascist regime, a period in which he had, of course, to avoid open criticism of the policy choices of the government but still "made his opinion clear in controlled critical comments" (Marchionatti 2000, p. xiv).

During his decades-long activity, Einaudi was in correspondence, and, in some cases, cooperated, with the most important economists of his time;⁸ suffice it to mention, among English-speaking economists, Edwin R. A. Seligman, Irving Fisher, Francis Y. Edgeworth, John Maynard Keynes, and Lionel Robbins. In 1915, in particular, at the invitation of Keynes, he wrote for the *Economic Journal* a detailed article on the state of public finance in Italy (Einaudi 1915e).⁹ In the early 1920s, Einaudi commented favorably in the *Corriere della Sera* on Keynes's analysis of the problems of the monetary stabilization in Europe. In 1933, on the contrary, he strongly criticized the policy proposals expressed by Keynes in *The Means to Prosperity* (Einaudi 1933a). In 1926, furthermore, Einaudi was appointed by the executive committee of the Laura Spelman Rockefeller Memorial as their representative in Italy for the nomination of candidates for fellowships in the United States and for supervising foreign Memorial fellows in Italy (Faucci 1986; Parisi 2010).¹⁰ Among these was Oskar Morgenstern, who in 1928 spent a semester of his fellowship visiting Italian academic institutions and who,

⁸ Einaudi's correspondence is stored at the archives of the Fondazione L. Einaudi in Turin. It includes 85,000 letters received by Einaudi and 6,000 written by him between 1894 and 1961.

⁹ A few years later, again at the urging of Keynes, he wrote a contribution on the postwar conditions of the Italian state budget for a special supplement of the *Manchester Guardian Commercial* (Einaudi 1922d; Marchionatti 2017). In 1923, furthermore, at the request of the Financial Committee of the League of Nations, Einaudi wrote, together with E. Seligman, Gijsbert Bruins of the University of Rotterdam, and Josiah Stamp, University of London, a report on the issue of the double taxation of income (Bruins, Einaudi, Seligman, and Stamp 1923).

¹⁰ In May and June 1926 Einaudi traveled to the United States at the invitation of the Rockefeller Memorial with the aim of gaining direct experience of the US institutions of higher education. On his tour he visited the

in the following years, was to maintain intellectual ties and friendship with Einaudi (Pavanelli and Nakayama 2008). As a leading representative of economic and political liberalism in Italy, Einaudi was also in contact with Austrian economists such as Friedrich Hayek and Ludwig von Mises, with William Rappard, director of the Graduate Institute of International Studies in Geneva, and with the German-born economist Wilhelm Röpke. In 1951, while still president of the Italian Republic (he held this position until 1955), he became a member of the Mont Pelerin Society.

III. EINAUDI'S ROLE AS OPINION MAKER

Einaudi started his work as a columnist in 1896, immediately after graduating at the University of Turin. In the previous years, the *Gazzetta Piemontese*, a daily newspaper founded in Turin in 1867, had undergone a radical transformation thanks to the resources provided by Alfredo Frassati, an ambitious journalist and businessman who had become the chief editor. Frassati renovated the equipment by purchasing a new rotary press, which ensured high printing runs, and renewed the newspaper by dedicating a page to cultural and literary themes and increasing the columns on economic and social issues. He reserved for himself the leading article, usually devoted to political issues. He was an advocate of social reforms and the extension of civil rights, and his pieces were characterized by a rough style and sharp criticism of the conservative political elite in government in Italy at the end of the 1890s, and found wide approval among the readers.

In the context of this process of transformation of the newspaper into a solid entrepreneurial initiative, a key step was to hire new collaborators: in particular, established writers, academicians able to intervene in an authoritative way on issues of interest to the public, but also promising young graduates. Among these was Einaudi, who at the very start was also required to contribute to the everyday work of the editorial office. With the new name of *La Stampa* (*Gazzetta Piemontese* became a subtitle), Frassati's newspaper—which is still published today—was bound to quickly become one of the most influential in Italy.¹¹

Between 1897 and 1902 Einaudi published an ever increasing number of articles in *La Stampa*: from thirty-two in 1897 up to 159 in 1901 (Bianchi and Giordano 2010, p. 46). However, in 1903 he decided to quit, probably in disagreement with the support manifested by Frassati to Giovanni Giolitti—the able but controversial political leader belonging to the liberal reformist camp who dominated the Italian political arena from 1901 till 1914 and promoted in 1912 the adoption of the universal male electoral suffrage but was severely criticized by many opponents for his often unscrupulous and authoritarian methods of government. ¹² During the same year Einaudi started to write for the

universities of Columbia, Yale, Harvard, Princeton, Minnesota, Berkeley, and St Louis, and had the opportunity to meet several leading economists of the time (Faucci 1986, p. 219).

¹¹ On the role played by *La Stampa* during this period, see Castronovo (1987, 2017). On the life and work of Alfredo Frassati (1868–1961), see Frassati (1978–82).

¹² Giovanni Giolitti (1842–1928) was Italy's most influential political leader between the late nineteenth century and the 1920s. A native of Piedmont, in northern Italy, he was elected to the Chamber of Deputies in 1882 and became prime minister in 1892. The following year, however, he was forced to resign because of his involvement in a financial scandal that led to the liquidation of a major bank of issue, the Banca Romana. He returned to lead the government in 1903 and continued to direct Italian politics until the First World War.

Corriere della Sera, published in Milan. This newspaper, founded in 1876, had gained a new momentum after the turn of the century when Luigi Albertini, a brilliant journalist and manager, emerged as its new chief editor. Having understood, during a stay in London, the potential of the daily press as a new independent industry capable of spreading new ideas and influencing public opinion but also of distributing dividends, he proceeded, as did Frassati, to entirely renovate the publishing activities. ¹³ It is worth mentioning that Albertini, who had graduated in law at the University of Turin, had been a member of the Laboratorio mentioned above, where he became acquainted with Einaudi. Thanks to relentless efforts and substantial investments in new rotary presses and technological modernization (systematic use of linotypes and telephone services), sales increased dramatically in the following years (from 76,000 copies a day in 1900 to 350,000 in 1912, and about 600,000 in 1917–18). ¹⁴ The increased revenues enabled Albertini to distribute high dividends to the shareholders of the limited partnership that owned the *Corriere*¹⁵ and to provide the newspaper's contributors with adequate pay. Consequently, he was able to attract some of the best journalists and writers of his time

In this period (frequently defined by historians as the "Giolitti era"), he sought the support of the socialist party, very strong in the north of Italy, by promoting legislation favorable to the workers but also adopting patronage practices such as the granting of government contracts to cooperatives linked to the trade unions. He also refused to use the army to repress workers' strikes, a practice adopted by previous governments. In the south of Italy, however, he often supported corrupt local cliques to the point of being defined by the historian Gaetano Salvemini as "the Minister of the underworld." His most important political achievement was the introduction in 1912 of the universal male suffrage. This reform, however, ultimately weakened the bloc of moderate political forces on which his power was based. In 1914–15 he opposed Italy's entry to the war. In June 1920, in a period characterized by severe economic difficulties and social tensions, he was again appointed by the king to lead the government but resigned the following year after obtaining disappointing results at the general elections (see Mack Smith 1997; Duggan 2008).

¹³ Luigi Albertini (1871–1941) was born in Ancona, central Italy, into a wealthy family (his father was a private banker). In 1889 the bank went into liquidation and the family moved to Turin, where Albertini graduated brilliantly in law with a dissertation on labor economics. In December 1894 he went to London to continue his research with the aim to start an academic career. In London Albertini came into contact with the administrative director of *The Times*, an acquaintanceship that gave him an inside view of the organizational and managerial complexities of a major newspaper. When he returned to Italy, he decided to devote himself full-time to journalism, also to meet his family's economic needs. Hired at the Corriere della Sera, he became the main collaborator of Eugenio Torelli-Viollier, the founder and director of this newspaper. After Torelli-Viollier's death, in 1900, Albertini took on the role of chief editor, consolidating the newspaper editorial line in favor of a modern liberalism open to social change. In 1915 Albertini, who in the previous year had become a member of the Italian Senate, engaged the Corriere in an active campaign in favor of Italy's intervention in the First World War. From 1923 he became a resolute opponent of Mussolini's government, writing several critical articles from the columns of the Corriere and taking the floor in Parliament to defend constitutional rights and the freedom of speech. As a consequence of that, in November 1925 he was forced by the fascist regime to give up his ownership shares and the control of the Corriere and to retire (Barié 1972; Moroni 2005; Romani 2012a).

¹⁴ See Benadusi (2011, p. 14).

¹⁵ The majority shareholder was Benigno Crespi, a cotton industrialist who chose not to interfere with the newspaper's editorial line. Significant shares were held by Ernesto De Angeli, also a textile industrialist, and by Giovanni Battista Pirelli, the founder of the company with the same name that specialized in the production of rubber goods. In 1900 the partnership's charter was revised to allow Luigi Albertini to join the shareholders, albeit initially with a tiny share. In 1919 Pirelli sold its share to Albertini, who became the second biggest partner after Crespi (Barié 1972, pp. 89, 510).

(among them, Einaudi).¹⁶ Thanks to his energetic leadership and ability to generate profits, Albertini had full control of the editorial policy. The *Corriere*, writes David Forgacs in a monograph on the formation of the cultural industry in Italy, "did not narrowly articulate the economic interests of its shareholders." Overall, however, Albertini's newspaper "was consistent with the values of the Lombard industrial bourgeoisie" (Forgacs 1990, p. 36).

Between the eve and aftermath of the First World War, the *Corriere della Sera* had become by far the most influential paper in Italy and one of the most authoritative in Europe. ¹⁷

As already mentioned, Einaudi's commitment to journalism predated his academic career and proceeded in parallel with it. ¹⁸ What matters most, he appeared to be fully aware of the crucial role played by newspapers as a key instrument in shaping public opinion. In a text written in 1928 but published several years later, he remarked that the daily press was the only information source for the majority of the literate Italian population during the second half of the nineteenth century; actually, the only channel through which "the ideas of philosophers and scientists" reached the public, and influenced politicians and compelled them to enact specific measures (Einaudi 1945b, p. 194). ¹⁹

¹⁶ Einaudi's pay was based on the number of articles he wrote and varied according to the length of the article and whether it was signed or not. In 1903 he was compensated for every article an average of 40 lira, and since 1910, 100 lira (Faucci 1986, p. 59). Applying the suitable coefficients of transformation, 100 lira of 1910 are equivalent to approximately 500 current US dollars.

¹⁷ See Barié (1972); Colarizi (2011). In his *Quaderni del carcere* (Prison notebooks), the philosopher and political leader Antonio Gramsci underlined the crucial role played by the *Corriere della Sera* in the Italian political arena in the 1910s and 1920s, writing that this newspaper had been a staunch advocate of the interests of the industrialists in the North as well as of several "enlightened" agricultural entrepreneurs in the South (Gramsci 1975, p. 2040). See also the following passage, published in *The Economist* in 1926, a few months after Albertini's forced removal as editor of the *Corriere*: "We cannot omit to remark upon the almost complete suppression of the freedom of the Press [in Italy]—which, not long ago, was startlingly brought home by the sudden change in the ownership and control of the *Corriere della Sera*. The *Corriere* was a great European as well as a great Italian newspaper, and its voice was one of the few remaining expressions of Italian freedom. Its acquisition by the Fascists in virtue of a legal technicality has lowered their prestige, while inflicting on Italy and Europe a signal disservice" ("The Progress of Fascism," *The Economist*, January 9, 1926, p. 56).

¹⁸ A revealing, although somewhat rhetorical, testimony of the importance Einaudi gave his work as opinion maker in the press and gave to the key role of Albertini is his letter of resignation from Corriere, which he wrote in November 1925 to Pietro Croci, the new chief editor imposed by the fascist government. "If in my soul there was the spark of that flame thanks to which one feels the high priest of an idea when writing; that spark could not have lit the flame if Luigi Albertini had not persuaded me, with his constant stimulation and the ardor of his enthusiasm that the journalistic priesthood was as noble and elevated as the academic priesthood" (Letter of L. Einaudi to P. Croci, Turin, Nov. 28, 1925, in Romani 2012b, vol. 2, p. 2234). On Einaudi's work as columnist on the daily press, see also De Cecco (2004); Romani (2012a); Berta (2012). ¹⁹ These reflections were expressed by Einaudi in a critical essay on Benedetto Croce's book *Storia d'Italia* dal 1861 al 1915 (History of Italy from 1861 to 1915). The text published in La Riforma Sociale, September 1928 (Einaudi 1928), however, did not include the section on the role of the daily press, which was deleted to avoid fascist censorship. The review in its entirety was instead included in a collection of essays printed in a limited edition in 1933 (Einaudi 1933b, pp. 125–151). After the fall of fascism, the section on the daily press was published in La Nuova Antologia, a quarterly journal published in Florence, together with an article written by Einaudi himself for the US magazine Foreign Affairs, and published in this journal in April 1945 (Einaudi 1945a and 1945b).

In the same article he emphasized the fact that, starting from the end of the nineteenth century, information in Italy had been greatly transformed by a few national independent newspapers (this was indeed the case for *La Stampa* and the *Corriere della Sera*, although Einaudi does not mention them explicitly). Unlike the traditional daily press funded by pressure groups and political parties, these newspapers had been managed as a business by a few entrepreneurs who had realized that in modern societies, the demand for information was increasing and they could gain substantial profits by providing this information in an efficient, reliable, and appealing way.²⁰

As a matter of fact, while the press that was directly linked to pressure groups accumulated losses and was unable to enlarge its readership, the newly created independent newspapers increased copies sold and, as stated, distributed substantial dividends to their shareholders.

This sound business foundation had become the basis for a "new social and political power" (Einaudi 1945b, p. 197). In other words, independent newspapers had become a driving force in shaping public opinion.

In his 1928 essay, Einaudi made an interesting distinction between the "legal" representation of interests in Parliament and "public opinion." The latter was not organized but, in his view, derived its strength from a fundamental need of the citizens to have free and non-partisan debate on the fundamental issues facing the country.

Public opinion, however, had to be "enlightened": "It is necessary," he wrote bluntly in 1905, "to fulfil the duty of informers and organizers of a healthy and vigilant public opinion" (Einaudi 1905c, p. 1). In this regard, a key point arises. Einaudi understood clearly that one of the crucial challenges facing Italy from the second half of the nineteenth century was its transformation from a country governed by a liberal elite on average intellectually capable and determined to complete the process of national unification but the expression of only very restricted social groups (only 3% of the population had the right to vote in the early 1860s)—into a modern democracy capable of allowing an active participation in political life of the middle class and of the workers while at the same time guaranteeing the full respect for the constitutional rights (including property rights) typical of a liberal system. In the years preceding the First World War, indeed, important steps in this direction were taken by Prime Minister Giovanni Giolitti: he had sought an agreement with the socialists and had promoted universal suffrage. Einaudi, as mentioned, was a strenuous opponent of Giolitti. In an article published in 1941 (reprinted in his collection of essays in 1954), however, he acknowledged that the latter had to be credited, at least, with "one great idea": that of "enabling the working classes and the peasants to be part of the political and economic government of the country." In implementing this program, however, wrote Einaudi, Giolitti had limited himself "to letting the socialists do the experiment." This was "too

²⁰ "If the manufacture of textiles, of machines," Einaudi wrote in this regard in the above mentioned essay, "requires miracles of intelligence, of organization, of cooperation of thousands of people, one dependent on the other and working for a common purpose; it is not to say what treasures of intelligence, of readiness to make decisions, of wise and elastic organization required the manufacture of that very delicate, intangible product which is the reliable news.... Italy admirably performed that task and performed it at such a low cost that it appears miraculous. Those who ... compare the great Italian newspapers of the time to *The Times* of London, the *Temps* in Paris, the German *Frankfurter Zeitung*, the *New York Times* or the *Chicago Tribune*, must conclude that our newspapers were second to none" (Einaudi 1945b, pp. 196–197).

little for a statesman who must be able to channel and lead the social forces to which he opened access to power" (Einaudi 1954, p. 286).

Einaudi's work as opinion maker and columnist appears, then, strictly connected with this complex process of transformation of Italian society and institutions and aimed at broadening the basis of consensus of economic and political liberalism in Italy.

In order to promote a widespread movement inspired by the principles of liberalism, indeed, monopolies, collusive practices, and unproductive rents had to be singled out and castigated. To defend savings and "honestly productive capital," Einaudi wrote in a letter to Alberto Geisser, a co-editor of *La Riforma Sociale*, he was ready to attack corrupt politicians and "take by the collar the capitalists robbing other people's properties."²¹

On this point the collaboration with the *Corriere della Sera*, in itself a great opportunity as a channel of transmission of his ideas, was also a constraint. As we will see later, Albertini was resolute in avoiding involving the *Corriere* in press campaigns considered too radical or such as to alienate moderate readers.

In his role of opinion maker Einaudi was indeed a gifted writer, whose ability to persuade could hardly be matched. In dealing with complex issues, he tried first of all to convey to the reader relevant information, both qualitative and quantitative. Only after setting out the problem did he make his point of view clear, advocating a specific measure or criticizing a policy adopted by the government. Einaudi's journalistic style reflects his habit of careful perusal and interpretation of official papers and statistical data, a habit he had acquired during his early years as research assistant at the Laboratorio di economia politica of Turin. Even more decisively, it reflects also *The Economist*'s principles. Einaudi greatly admired the weekly journal founded by James Wilson in 1843 at the height of the anti-corn laws movement, and which had become, with Walter Bagehot, an influential source of information for the business community in Great Britain and elsewhere (Edwards 1993; Berta 2012). In particular, Einaudi appreciated *The Economist*'s commitment to the principles of free competition and individual initiative, and its determination to adopt them as a guideline, but also its pragmatism.

Between 1896 and 1925 Einaudi signed only a portion of the numerous articles he wrote for *La Stampa* and the *Corriere della Sera*: usually, but not necessarily, those in which he focused on key policy problems. At any rate, he soon acquired a large following among the readers. Umberto Ricci, professor of economics at the University of Rome, after praising him for teaching the principles of political economy to thousands of people, observed that he was increasingly considered by his readers as "an oracle" (Ricci 1917, p. 392). Over the years Einaudi received innumerable letters from private citizens or representatives of institutions who praised him for a specific position or denounced wrongdoing and inefficiencies.²²

Furthermore, as a leading columnist of the most influential newspaper in Italy, on several occasions Einaudi addressed Italian political leaders and ministers directly, criticizing them for not having taken into account his remarks in the columns of the

²² These letters are stored at the archives of Fondazione L. Einaudi, Turin.

²¹ Fondazione L. Einaudi Archives, *Geisser*, letter of L. Einaudi to A. Geisser, Dogliani, Oct. 4, 1910.

Corriere and urging them to act.²³ Not infrequently, his criticisms triggered worried replies from the politicians involved.

In his work at the Corriere, Einaudi cooperated strictly with the chief editor, Luigi Albertini, with whom he shared a strong commitment to the principles of liberalism and a common vision of the role and duties of the Italian bourgeoisie. As mentioned, however, Albertini was firm in his role as having ultimate responsibility for the continuity and coherence of the editorial policy of the newspaper he directed. In his letters to Einaudi, besides constantly stimulating and inspiring him on the subjects to tackle in his columns, he intervened several times to smooth out what he perceived as excessive or untimely criticism. In some cases, he did not hesitate to send one of Einaudi's articles back, urging him to revise it substantially or to publish it in another journal. Furthermore, at least until the First World War, Albertini was reluctant to criticize the Italian trade tariff, characterized by heavy protectionism, and vetoed any proposal to embark the *Corriere* on a free trade campaign. In this regard his views were in tune with those of Luigi Luzzatti, the influential protectionist economist and political leader who had been the architect of the Italian tariff and who frequently wrote in the Corriere on policy issues. Consequently, for several years Einaudi published his articles in favor of free trade and his denunciations of the rent-seeking activities of the steel or the oil industries²⁴ in his monthly journal, La Riforma Sociale, and in L'Unità, edited by Gaetano Salvemini, 25 both characterized by limited circulation.

As already mentioned, Einaudi was strongly committed to free competition and private enterprise. In his articles he sided with the numerous Italian entrepreneurs who were able to compete and to sell their products on the domestic and international markets without seeking subsidies or protective duties from the government, with the artisans and workers who thrived through hard work and self-restraint, and with the peasants who managed to acquire the land they worked thanks to lifelong savings. At the same time, he castigated the inefficiencies of the public sector, the increasing influence of the bureaucracy, and the irrationality of the Italian tax system. Indeed, particularly after the First World War, Einaudi's attitude appears to be that of a preacher whose sermons went often unheard but who had to provide them anyway, to fulfil a precise duty.²⁶

²³ "I deeply regret," he wrote, for example, in May 1914, "that the Minister did not take sufficiently into account the remarks made in these columns on the relationships between protectionism and trusts" (Einaudi 1914a, p. 1).

²⁴ The subsidized entrepreneurs of the oil industry were defined by Einaudi as "state drillers," a term he later used figuratively to define all economic operators who were able to survive only by "drilling" the state budget (Einaudi 1911a, 1911b). On Einaudi's cooperation with Albertini, see Faucci and Becchio (2010); Romani (2012a). On the free-trade movement in Italy from the end of the nineteenth century until the 1920s (which involved, in addition to Einaudi, Antonio De Viti de Marco, Attilio Cabiati, and Gaetano Salvemini) and on the anti-protectionist campaigns it promoted, see Allio (2000); Tedesco (2021).

²⁵ L'Unità was a weekly newspaper founded in 1911 by Gaetano Salvemini with a democratic and anti-protectionist program. It ceased publications in 1920. It is not to be confused with the political daily newspaper of the same name founded in 1924 by Antonio Gramsci.

²⁶ In 1920 Einaudi republished under the title *Prediche* (Sermons) several pieces he had written during and immediately after the First World War, in which he exhorted his fellow citizens to reduce consumption and to employ all available resources to the need of the conflict. "These writings," he observed in the introduction, "can be rightly called 'sermons' because, as it happens usually to the warnings of the economists, they

IV. EINAUDI'S COLUMNS ON ITALY'S ACHIEVEMENTS AND CONSTRAINTS

In his work as a columnist at *La Stampa* and the *Corriere*, Einaudi analyzed the main challenges facing Italian society of his time: the evolution of the industrial relations and the role of the trade unions; the industrialization of northern Italy; the backwardness of the agriculture in the South; the expansion of international trade; the rationalization of Italy's railways, ports, and navy transport; and the need to reform the public administration and to control public expenditure.

This section focuses on the main issues tackled by Einaudi, starting from his analysis of industrial relations, the often turbulent confrontation between the Italian workers and the industrialists. In a series of articles published in *La Stampa* in September and October 1897 and reprinted in his *Cronache economiche e politiche di un trentennio* (1959), Einaudi provided his readers with an accurate report of the first organized strikes in the textile sector in Piedmont, focusing on the main features of this industry, on the living conditions of the workers, and on their personal aspirations and expectations (Einaudi1959, pp. 40–62). Three years later, in December 1900, he took on the task of reporting on a major strike proclaimed in the port of Genoa after the disbandment, decreed by the government, of the local trade union controlled by the socialists. Once again, he did not limit himself to a description of the facts but analyzed the causes and the likely consequences of the confrontation (Einaudi 1959, pp. 290–309).

Einaudi's reflection on strikes is closely related to that on the nature and role of the first trade unions. Most of these organizations, he observed, had emerged spontaneously as a result of the efforts of individual workers who, animated by a legitimate desire to improve their living conditions, realized that it was better to advance their claims in a collective manner rather than at an individual level. It is not surprising, therefore, that Einaudi was against any legislation aimed at imposing a regimentation of the unions: his model was Great Britain, where, he wrote, there was "the maximum freedom for all, workers and entrepreneurs, to associate to defend their interests" and at the same time it was possible to withdraw from an association when it was no longer considered convenient to be part of it (Einaudi 1901, p. 1).

A lively confrontation in the labor market, in any case, maintained Einaudi, was a clear indicator of the fact that Italy was a growing and dynamic society. ²⁷ After a period of financial turmoil and recession during the 1890s, at the turn of the new century Italy was finally experiencing a substantial increase in industrial production, export, and income per capita. In his articles Einaudi ascribes this positive process to the creativity and determination of thousands of entrepreneurs, craftspeople, and workers, who had been able to face an increasing domestic demand for consumption goods and also to compete at an international level without requesting government subsidies or protection. The stabilization of the banking system and the adoption of a more responsible fiscal policy had also had a positive influence.

remained unheard; as a consequence, the economic and social costs of the war were by far superior to those ... which was reasonable to expect" (Einaudi 1920a, p. vii).

²⁷ For an overview of Italy's economic conditions in this period, see Zamagni (1993).

On the contrary, no specific merit was attributed to the Italian governments and particularly to Giovanni Giolitti, whom, as previously mentioned, Einaudi opposed as a politician ready to accept any compromise in order to stay in power.²⁸

According to Einaudi, a sign of Italy's increasing capacity to produce was the fact that its international trade had been steadily growing since the beginning of the new century. Italy, he observed in an article published in May 1905, imported from abroad a growing amount of raw materials and instrumental goods to create new finished products: "We are no longer a purely agricultural country ...; we are also a country which lives by its own industries and buys abroad raw material to be transformed" (Einaudi 1905a, p. 1). For the time being, exports were not actually increasing with the same vigor as imports; as a consequence, the balance of the trade was in deficit. Einaudi, however, was not particularly concerned, considering it a temporary imbalance, bound to flatten out in the medium run.

Another relevant issue in those years was a huge increase of emigration, particularly from southern Italy. In the 1890s the average number of emigrants had been 300,000 per year and from 1909 to 1913 the outflow increased to 700,000 (Candeloro 1981, pp. 131– 132). Einaudi dedicated some of his most eloquent pieces in La Stampa to this mass exodus, describing at length the sufferings of emigrants—often victims of abuse perpetrated by local authorities and speculators—but also envisioning the birth of a "great transatlantic Italy" in Latin America, thanks to the labor and creativity of the Italian citizens abroad (Einaudi 1959, pp. xii, 350–367). In the *Corriere* he returned to this subject with some significant articles, in which his concern about the negative consequences of an emigration of such a large scale on Italy's human capital, while significant, recedes into the background compared with his appreciation of its positive aspects as a security valve against the excess of labor supply in the domestic market. The solution, in his view, was to promote economic growth in the South through new infrastructures, a rational use of water resources, and land improvement (Einaudi 1906g). Moreover, several emigrants tended to return, after a period of work abroad, to their homeland, bringing new skills and capital, albeit modest, which they used to acquire land property (Einaudi 1910).

One of the most recurring themes tackled by Einaudi at the *Corriere* was the need to make public administration and finance leaner and more efficient. Thanks to substantial restructuring efforts accomplished in the previous years and also to the economic upturn, the government budget at the turn of the twentieth century was in much better condition than in the 1890s. The trustworthiness of the country, which had been gravely jeopardized in that period by a wave of financial scandals and by excessive military expenditures in the colonies, was now re-established, and in 1906 the government succeeded in obtaining a remarkable reduction of the interest rate of the consol, by far the most important government bond at the time, thus relieving the impact of interest payments on

²⁸ See Einaudi (1911c, p. 884; 1906a). Einaudi's criticism became particularly harsh on the eve of the First World War: Giolitti tried to avoid Italy's participation in the war, but Einaudi and the *Corriere* campaigned vigorously in favor of it. From 1920 to 1922, however, the criticism subsided, and Einaudi praised Giolitti's ability to reduce Italy's budget deficit.

the state budget.²⁹ However, numerous issues remained open, starting with the need to implement strict rules in public expenditure and to reform Italy's obsolete and inefficient tax system.

In Einaudi's opinion, the former issue was the most pressing one, in order to ease the burden on Italian taxpayers and to enhance economic growth. Instead, the government was heading in the opposite direction: in particular, it was continually inflating the number of employees in the public sector for electoral purposes and under the pressure of the bureaucrats.³⁰

The need to defend the country's financial stability against the demands of pressure groups to the detriment of the Treasury dictated to Einaudi the content of some of his most effective pieces. In several articles, for example, he criticized the complex system of maritime conventions stipulated by government in collaboration with private shipping companies. These agreements were admittedly crucial, in many cases, to guarantee postal services and vital connections to remote coastal areas of the peninsula and to the colonies. However, if badly structured, they risked causing excessive burdens for the taxpayers (Einaudi 1906b, 1906c, 1906d).

In spite of the abovementioned problems, the trend of the Italian economy before the First World War was remarkably positive. It was, however, accompanied by fluctuations in the financial markets, the most serious being the 1907 crisis (Bonelli 1971; Zamagni 1993, p. 181). These complex issues were tackled by Einaudi in some interesting articles.

Starting from 1901, the average price of stocks at the Genoa, Milan, and Turin stock exchanges had been dramatically increasing. The stock market reached an unprecedented peak in the car industry, located in the northwestern part of the country and still in its early stages. In an article published in June 1905, Einaudi stigmatized this event as a speculative "fever," similar to other episodes that had occurred in the past. Broadly speaking, the fact that Italian households were putting their savings in the stock market was in itself positive: had they not been excoriated in the past for having been too conservative and investing too much in government bonds? Investors in the stock market had to be prudent nonetheless, distinguishing between already consolidated sectors and "new industries," in which profits had not been reduced to a normal level through competition. The car industry was one of the latter and undoubtedly had been affected by speculative maneuvers. A part of the blame had to be put on the banks, who should have "separated the wheat from the tares": in other words, they should have recommended the more solid assets over the less reliable ones (Einaudi 1905b).

The bull market was bound to continue until October 1906, when the Genoa Stock Exchange nearly collapsed as a consequence of a major liquidity crisis triggered by an ill-conceived speculation on the stocks of Società Terni—a major steel company that specialized in armor for the Italian navy—by its president, Ferruccio Prina (see Bonelli 1971; Tusset 2016).

Solicited by Albertini, Einaudi dedicated to this theme an extensive article published by the *Corriere* at the beginning of November with a special editorial introduction. In the

²⁹ As a consequence of this measure, denominated "conversione della Rendita" (conversion of consolidated bonds), the nominal interest rate paid on that bond was reduced from 4% to 3.5%. See Zamagni (1993, p. 176).

³⁰ This issue is frequently reasserted by Einaudi in his articles in the *Corriere*; see Einaudi (1960, pp. 316–327).

last few years, wrote Einaudi, following the creation of many new industrial enterprises, the market had been inundated by a huge number of stocks. The banks, who had initially been ready to accept them as collateral, became more cautious and, in several cases, traders who had incurred a substantial debt were forced to "distress selling." Another cause of the fall in stock price was the worldwide increase in interest rates. In the light of all this, speaking of "bearish plots" or conspiracy by speculators was pointless. On the whole, stock markets had to be considered a key institution for the economy (Einaudi 1906e).

Einaudi's article aroused great interest among the readers, and a few days later Albertini was writing to commission a further article on stock speculation focusing on Prina's case with Società Terni. In the same letter Albertini specified the topics to be treated. Einaudi, however, followed only in part the suggestions of the chief editor of the *Corriere*, transforming his piece into an act of accusation against the steel industry, which at that time was heavily subsidized by the government. Following Albertini's wishes (which were also in line with his views), he argued that corporate managers should abstain from stock speculation and in particular from attempting to artificially manipulate the price of company stocks. Their only duty was, and should have been, to adopt adequate entrepreneurial strategies and to put aside part of the profits in order to reinforce the corporate reserves. Having dealt with this issue, Einaudi did not hesitate to openly criticize the Terni management: the stock price collapse was partly a consequence of the fact that the government had at last remembered its duty towards the taxpayers and had started questioning the inflated contractual conditions of the steel sector (Einaudi 1906f).

The months of October and November 1907 witnessed a worsening of the crisis at an international level. This time the epicenter was in the United States: at the New York Exchange the stocks hit bottom. In an article published on November 18 ("The American Tornado"), Einaudi identified with his usual clarity the factors provoking the overseas crisis, which had followed a long period of euphoria and boom.³²

In the following years, on the eve of the World War I, the nominal interest rates increased on the financial markets. This led to a reduction in the price of government bonds and particularly of the consols, in Italy and abroad. In an article published in February 1912, Einaudi, endorsing explicitly Irving Fisher's view, attributed the rise of nominal interest rate to the increase of the rate of inflation: "the scientific explanation," he wrote, "[had been] given by Fisher, no doubt the greatest living American economist, an insightful observer of the current situation" (Einaudi 1912a, p. 4). In his article he also mentioned Fisher's distinction between the nominal and the real interest rate, an analytical point that at that time was quite innovative.

Therefore, in order to counteract the price reduction of the government bonds, it was necessary to stabilize the cost of living. In another article, published a few days later, Einaudi explicitly mentioned the stabilization plan set forth by Fisher in his *Purchasing*

³¹ Letter of L. Albertini to L. Einaudi, Milan, Nov. 18, 1906, in Romani (2012b, vol. II, pp. 1003–1005). ³² Starting from 1903, wrote Einaudi, the US economy had experienced an unprecedented recovery, which had fostered investment projects exceeding the availability of saving. This had fueled speculation, which had been encouraged by banks and newly created trusts aiming at raising prices artificially. As soon as the price of stock dropped, this "house of cards" collapsed (Einaudi 1907). On Einaudi's analysis of the 1907 crisis and the conditions of the Italian economy in this period, see Cassata and Marchionatti (2010).

Power of Money. Fisher's plan, observed Einaudi, was convincing, at least in principle, even though not all economists agreed with its theoretical assumptions and particularly with the hypothesis that the increase in prices had been caused by an excess supply of gold (Einaudi 1912b).³³

V. REDUCING CONSUMPTION TO FINANCE THE WAR

One of the worst consequences of the First World War was, undoubtedly, the collapse of the complex network of institutions and rules—mainly informal—that, by fostering the free movement of consumption goods, raw materials, and factors of production, was at the very basis of the unprecedented increase in income and wealth that had characterized Europe and North America from the second half of the nineteenth century.

Einaudi was aware of the key role played by this system and indeed he repeatedly defended it against criticism. Nevertheless, from the end of 1914 he actively campaigned, together with Albertini, for Italy's intervention in the conflict against Austria and, later, against Germany. Similarly to other intellectuals of his generation, he believed that the war would have led to the completion of the national unification process initiated in the second half of the previous century, reinforcing at the same time the position of Italy as one of the leading political and military powers at an international level. Furthermore, by aligning with France and the United Kingdom, Italy would have contributed to defending the liberal values that appeared to be threatened by the German bloc.³⁴ As demonstrated by the events in the following years, this analysis actually proved to be a tragic underestimation both of the human and economic costs of the conflict and of its political and social effects. In Italy and in other belligerent countries, the liberal institutions and constitutional rights were ultimately brought to a collapse by the disruptions of the war and the irrational expectations and myths it created among the population.

The articles written by Einaudi during the war period were characterized by an austere and patriotic tone. The primary objective was to reduce private consumption to the minimum, if possible through moral suasion or, if necessary, through the law (see, among others, Einaudi 1916a, 1916b). Public investments in infrastructure had to be postponed, too. The resulting resources should be raised through taxation (extra profits deriving from army supply contracts should have to be specifically targeted) or through the purchase of government bonds by the households and then used to finance the war. Only in this way would it have been possible to avoid a huge increase in paper money, which would have fueled inflation and caused iniquitous redistributions of income.³⁵

Increasing tax revenue, though, was quite a slow process. Therefore, in the immediate future the purchase of government bonds by Italian households was the best way to pay for the war without hampering price stability. As a matter of fact, in the years 1915 to

³³ On the reception of Irving Fisher's analytical work and policy proposals in Italy before and after World War I, see Pavanelli (2006).

³⁴ On this point, see Vivarelli (1981, pp. 284–291). Italy entered the war against the Austro-Hungarian empire in May 1915 and against Germany in August 1916.

³⁵ Einaudi was one of the few Italian economists who believed that, at least in principle, the costs of the war could be covered "without resorting to the inflation tax" (Fratianni and Spinelli 1997, p. 113).

1918, five major issues of long-term bonds were launched by the government (Fausto 1993, pp. 3–7; Fratianni and Spinelli 1997, pp. 114–117).

In his articles in the *Corriere*, Einaudi actively engaged in a systematic work of persuasion and propaganda, urging people to reduce consumption and to subscribe wholeheartedly to the massive issuance of bonds by the Treasury. Referring to the first issue of one billion lira launched in January 1915, when Italy was not yet taking part in the conflict, Einaudi argued that by buying bonds the savers were not only fulfilling a patriotic duty but also acting according to their best interest, as the conditions offered were very favorable (a forecast that, of course, proved to be wrong in that it underestimated the effects of inflation on the real interest rate and on the value of bonds; see Einaudi 1914b and 1915a).

Commenting on the second issue of bonds, promoted in June 1915, after having emphasized the fervor pervading the country to the point that many citizens were ready to incur debts in order to acquire these assets, he stigmatized the lack of commitment of most industrialists despite the fact that they had accumulated large profits thanks to military contracts (Einaudi 1915b).

In the first two government loans, wrote Einaudi in another article, the Italian savers had provided as a whole two billion lira compared with fifteen billion in Germany and twenty-four billion in England. In normal conditions, this result would have been acceptable, keeping in mind the big disparity between Italy's income and that of other European countries. Italy, however, was taking part in a major war: in this context the resources collected were not enough to avert the danger of forced loans and inflation. Politicians, lamented Einaudi, had not been very active in promoting the loan in their constituency. The intellectuals, and particularly the economists, had also been, with few exceptions, conspicuously absent. This was not acceptable: there were circumstances in the life of a nation, he wrote, in which intellectuals should try to keep their critical attitude silent and "act as preachers" (Einaudi 1915c).

Einaudi himself, it must be added, was not completely consistent with this guideline. In December 1915, in agreement with Albertini, in the *Corriere* he criticized some clauses of a new issue of government bonds that determined an unjustifiable difference of treatment among subscribers.³⁶ On the whole, however, particularly after the severe defeat endured by the Italian army in Caporetto³⁷ in October 1917, his propagandist zeal on this issue prevailed over scholarly analysis. In this difficult period, when inflation increased and Italian households faced the risk of losing, in real terms, most of the capital invested in government bonds, he continued to urge his readers to buy these assets by presenting this investment decision as a bargain as well as a patriotic duty.³⁸

Several articles published by Einaudi in the *Corriere* during this period provided the documentary materials for a major book published in 1933 in the Italian series of the

³⁶ Einaudi (1915d). On this point, see Pavanelli (2012, pp. clix–clxii).

³⁷ The battle of Caporetto was fought at the end of October 1917, on an area near present-day Slovenia, between the Austrian and German armies on one side and the Italian army on the other. It marked a major defeat for the Italian army, which had to retreat on a new defensive line (along the river Piave). As a consequence, a new government of national unity was formed that proceeded to reorganize the armed forces. ³⁸ "After the war ... the value of the Treasury bills will increase. Experience shows that this always happened in the past and there is no reason to doubt that this will happen now.... The price of the corn, quite the contrary, will go down" (Einaudi 1918a, p. 2; see also Einaudi 1917a and 1918b).

Carnegie Endowment's economic and social history of the war, and which is still one of the most insightful works on Italy's economic and social events in that period: *La condotta economica e gli effetti sociali della guerra italiana* (The economic conduct and the social effects of the Italian war) (Einaudi 1933c).³⁹

VI. THE RECONSTRUCTION PROGRAM

In the aftermath of the armistice, proclaimed in November 1918, Italy had to face economic and social problems on an unprecedented scale. In order to cope with the needs of the war in a context in which revenues from taxes were by no means sufficient, the Italian government had borrowed huge resources both from domestic households (by issuing an increasing amount of bonds) and from abroad (mainly raw materials and military equipment from the United States and Great Britain). Further financial resources had been provided by the Bank of Italy. The ensuing increases in paper money, as was predictable, fuelled inflation.

Besides all this, it was necessary to promote a complex reorganization of the industrial sector, which had been entirely mobilized for the needs of the war and now had to convert to original production.

More generally, the experience of war brought new hopes of radical social and political changes. These aspirations, repressed during the conflict, were bound to break out in the years of 1919 and 1920 as a wave of strikes and street protest. In this dramatic and complex situation, Einaudi reacted by increasing his efforts as opinion maker at the *Corriere*. In his articles published in this period, he was able to devise a national "reconstruction program" based on fiscal restraint, monetary stability, and the defence of free competition and market economy (Barié 1972, pp. 389–390).

In Einaudi's view, balancing the government's budget was indeed the most pressing problem of the day. This goal had to be reached, on the one hand, by introducing a radical reform of the Italian tax system and, on the other, by drastically reducing public spending. Public subsidies, particularly those aimed at keeping the price of bread artificially low, had to be abolished. The consolidation of public finances, in turn, was a prerequisite for a freeze of the Bank of Italy's liquidity creation and for price stabilization. According to Einaudi, domestic price stability would have halted the depreciation of the lira and led soon to exchange rate stability. This was the precondition to the adoption, in due time, of the gold standard. Contrary to other economists of his time, however, he opposed pegging the lira to gold too hastily: a reasonable period of a de facto stabilization was needed. Even more so, he was against restoring the pre-war parity of the lira with the pound and the dollar, as such a measure would have led to an unsustainable deflationary process.⁴⁰ On this point, Einaudi commented favorably in the

³⁹ Two others of Einaudi's publications include articles written during the war and postwar period: *Prediche* (Sermons) (Einaudi 1920a) and *Gli ideali di un economista* (The ideals of an economist) (Einaudi 1921). ⁴⁰ Einaudi (1923a). Between 1913 and 1921 the value of the lira in terms of the dollar decreased from 5.27 to 23.46; in terms of the pound, from 25.71 to 90.17. See Ciocca and Ulizzi (1990).

Corriere on an article by Keynes, also published in Albertini's newspaper on April 29, 1922.⁴¹

Fiscal restraint and price stabilization were, of course, not enough. A further key issue was the need to dismantle the complex and inefficient system set up during the war to control economic activity and channel it towards military production: a system that was based on arbitrary rules and restrictions. In this framework, it was crucial to get rid of top bureaucrats who were convinced they had superior knowledge and let businessmen produce and sell freely without special authorizations, extra taxes, and pleas (Einaudi 1919a). It was also necessary to abolish the exchange rate control established in 1917, which created enormous obstacles to the purchase of raw materials and manufactured goods essential to the relaunching of national productions (Einaudi 1919b).

The government, he wrote, should choose between "regulated and free market economy, between those who require a State-led system and those who maintain that the State should only set the framework within which private business can thrive" (Einaudi 1920c, p. 1).

Once the major obstacles to economic activity had been removed, however, entrepreneurs should pull their sleeves up and avoid asking the government for new favors. On this subject, Einaudi strongly criticized the requests, made by the main business associations, to substantially increase custom duties on imported manufactured goods. Contrary to what had happened before World War I, when he had to soften his stance on this issue, he was now able to campaign vigorously against protectionism in the columns in the *Corriere della Sera*, accusing those who favored the policy of fuelling "class selfishness" and slandering Italian farmers and industrialists by representing them as people unable to succeed without constant government protection (Einaudi 1919c).

Another important issue in Einaudi's articles was the radicalization of the social and political conflicts in Italy. In 1920 the CGL, the trade union linked to the Socialist party, launched a wave of strikes during which factory committees of unionized workers tried to take control of several major industrial plants in the Northwest, exonerating the managers and the owners. In September many plants were occupied by the workers in Turin and Milan. Several weeks of strong tension ensued, during which a violent uprising inspired by the Soviet revolution appeared imminent to several observers (Berta 1995, pp. 16–17).

Writing before these dramatic events, Einaudi conceded that factory committees could play a useful role, provided that they limited themselves to discussing wages and working conditions in the firms. However, he strongly condemned the occupation of the industrial plants and the objective of "self-ruled management" pursued by the workers.

⁴¹ See Einaudi (1922b); Keynes (1922). In April 1922 the issue of the restoration of the gold standard was addressed at an international conference held in Genoa. Keynes covered the conference for the *Manchester Guardian* and wrote a few articles on this subject, syndicated to several European newspapers, including the *Corriere* (Skidelsky, 1992, pp. 106–111). In a letter dated April 30, Keynes wrote to Einaudi: "I was most delighted to read your article in yesterday's *Corriere* commenting on myself.... Although I am a little more optimistic than you are about the possibilities of the situation, there is very little difference between our opinions. We both agree that the immediate problem is to persuade the world that a return to the pre-war parity is an absurdity" (Letter of J. M. Keynes to L. Einaudi, Santa Margherita, Genoa, April 30, 1922, quoted in Marchionatti 2017, pp. 274–275). In 1920 Einaudi had published in *Corriere* a laudatory review of Keynes's *The Economic Consequences of the Peace* and, two years later, a positive comment on *A Revision of the Treaty* (Einaudi 1920b, 1922a).

These were disgraceful experiments, he wrote, which, besides violating the pre-existing property rights, were based on the wrong assumption that the control of the "capital" was a sufficient condition to create a new and even more efficient economic system. On the contrary, the most likely result would be misery and chaos (Einaudi 1920d).

The acute social confrontation and the perspective of a violent overturning of the political and economic system induced Einaudi to harden his opposition towards the socialists. At the same time, he failed to criticize the violence perpetrated by the first fascist "action squads" and, in September 1922, he even depicted fascism as a movement that drew inspiration from the "old liberal tradition" (Einaudi 1922c). This was, of course, a tragic misunderstanding, which indeed was shared at that time by several Italian intellectuals.

After Mussolini's appointment as prime minister in October 1922 and the formation of the so-called national government, Einaudi evaluated positively the work of the new minister of finance, the economist Alberto de' Stefani, praising him for restoring fiscal discipline and for enforcing a credible plan aimed at balancing the budget.⁴² During 1923, however, he expressed an increasing unease towards the authoritarian methods adopted by Mussolini and particularly his strategy to marginalize the role of Parliament. This criticism became an overt opposition after the assassination, perpetrated by a fascist squad, of one of the leading members of the opposition, the moderate socialist Giacomo Matteotti. In an article published in the *Corriere* on August 6, 1924, Einaudi denounced the spread of illegality and violence in the country and strongly condemned the government's attempts to abolish constitutional rights, particularly the freedom of the press. At the same time he urged the industrialists, most of whom, he wrote, were able to prosper and thrive without depending on public subsidies or favors, to speak up and to condemn these abuses (Einaudi 1924).

This passionate, although belated, plea was in vain: in the following months Mussolini was able to consolidate his dictatorship and to silence all opposing views. In November 1925 Luigi Albertini was forced to give up the leadership of the *Corriere* and Einaudi decided to suspend any collaboration with the newspaper, now controlled by the regime.⁴³

From the end of 1925 a long period began for Einaudi of "internal exile," characterized by a forced retreat to academic teaching and research, although he still tried to defend his ideals, writing in *La Riforma Sociale* and, when this was suppressed by the regime in 1935, in a new journal, *Rivista di Storia Economica*. The latter being—at least officially—specialized in economic history, could bypass, until the early 1940s, the rigid censorship of the regime. In this difficult period Einaudi probably came to the conclusion

⁴² The achievement of this goal, however, Einaudi wrote, had been greatly facilitated by a few unpopular measures adopted in 1920 by Giolitti, then prime minister, particularly his decision to abolish a generous state subsidy aimed at keeping the price of bread artificially low (Einaudi 1923b). On the policy measures adopted by Alberto de' Stefani as minister of finance from November 1922 to June 1925, see Zamagni (1993, pp. 244–248); Pavanelli and Bianchi (2020, pp. 149–151).

⁴³ In a letter to Albertini dated November 20, Einaudi reaffirmed his intellectual affinity with Albertini and reiterated his opposition to write for the *Corriere* once it had passed into other hands: "There is no other person in Italy with whom I could feel in communion of thoughts and the so-called 'technical' collaboration is deeply repugnant to me. ... One cannot and should not technically collaborate with people with whom one does not share ideals and feelings" (Letter of L. Einaudi to L. Albertini, Turin, Nov. 20, 1925, in Romani 2012b, pp. 2231–2233).

that his activity as preacher had been almost useless. However, he wrote in the 1950s after the re-establishment of constitutional rights in Italy: "To those who have in their blood the imperative to write it serves no purpose to be persuaded of the futility of their work.... As soon as I recovered the freedom of putting on paper my own thought, the habit revived and I was forced again to write" (Einaudi 1956, p. ii)

As a matter of fact, on July 28, 1943, just three days after the fall of Mussolini's government, he wrote to Ivanoe Bonomi, an anti-fascist political leader who in 1944 became prime minister of the Italian democratic provisional government, declaring himself "ready to resume after 18 years, since November 1925, my collaboration with the daily press." 44

VII. CONCLUDING REMARKS

In retrospect, Einaudi's work as columnist in the daily press and particularly his collaboration with Luigi Albertini's *Corriere della Sera*, by far the most influential and widely read Italian newspaper in the first decades of the twentieth century, appears to be a unique chapter in the diffusion of economic and political liberalism in Italy, an ambitious attempt to broaden the consensus in public opinion on the principles of free market and competition—then shared only by a minority of the population—in a context of a difficult process of transformation of the country in a modern democracy open to the participation in political life of new social classes.

The economic discourse conveyed by Einaudi in his pieces for the *Corriere* was certainly not "value free": he praised thrift, frugality, and individual responsibility as the only way for workers and artisans to improve their conditions and for businessmen to thrive; sided with the small entrepreneurs who were able to compete and to succeed without claiming subsidies and public aid; and condemned as inefficient per se any attempt by the government to coordinate economic activity or to promote specific sectors.

In his decades-long intellectual activity, Einaudi abhorred the tendency to interpret the role of the economist as pure "technician." In one of his last contributions, presented at the Mont Pelerin congress held in Turin in September 1961, he reiterated the view that had guided him in his work, namely that economists should act as intellectuals able to grasp the relationship between economic actions and "the political or moral or spiritual action[s]" (Einaudi 1962, p. 260), and, also for that reason, ready to actively participate in policy debates.

Einaudi's work as opinion maker was indeed a chapter in many ways unique for the economic discourse in Italy and the spread of liberal ideas and values in public opinion. As such, it was not exempt from constraints and shortcomings: this is the case of the anti-protectionist battle or his battle against the rent-seeking activities of the steel and oil companies, indeed central issues of any program inspired to economic liberalism and which Einaudi could not deal with in the *Corriere*, at least until the first postwar period, because of Albertini's reluctance to engage his newspaper on campaigns that could alienate moderate readers. During the same years, on the whole, Einaudi paid limited

⁴⁴ Letter of Luigi Einaudi to Ivanoe Bonomi, Turin, July 28, 1943, quoted in Faucci (1986, p. 310).

attention to the problems of southern Italy, characterized by secular backwardness. Also, the strenuous support of Einaudi and Albertini for Italy's intervention in the First World War turned out to be a tragic underestimation of the costs of the conflict and of its consequences at political and social levels.

As a matter of fact, despite the efforts made in the first postwar period by Einaudi to promote in the columns of the *Corriere* a program of economic and social reconstruction of the country inspired by the principles of economic liberalism, the political system and constitutional rights in Italy would soon be overwhelmed by the myths and vested interests created by the war—myths and interests that were skilfully exploited by Mussolini and the fascist movement to conquer the existing power structure.

The defeat, however, proved not to be irreversible. Immediately after the collapse of Mussolini's regime, in July 1943, Einaudi was ready to assume again the role of "untiring educator" (Cafagna 2004) and preacher, which would have led him to collaborate again with the daily press and, fairly soon, to take on roles of political leadership in the new Republic.

COMPETING INTERESTS

The author declares no competing interests exist.

REFERENCES

- Allio, Renata. 2000. "I trivellatori di Stato. L'antiprotezionismo (1894–1914)." In C. Malandrino, ed. *Una rivista all'avanguardia. La 'Riforma Sociale' 1894–1935. Politica, società, istituzioni, economia, statistica.* Firenze: Olschki, pp. 315–342.
- Augello, Massimo M., and Marco E. L. Guidi, eds. 2001. *La scienza economica in Parlamento 1861–1922. Una storia dell'economia politica dell'Italia liberale*. Volume II. Milano: FrancoAngeli.
- ——. 2003. Gli economisti in Parlamento 1861–1922. Una storia dell'economia politica dell'Italia liberale. Volume I. Milano: FrancoAngeli.
- ——. 2005. Economists in Parliament in the Liberal Age (1848–1920). Aldershot: Ashgate.
- Augello, Massimo M., Marco E. L. Guidi, and Giovanni Pavanelli, eds. 2016. *Economia e opinione pubblica nell'Italia liberale. Gli economisti e la stampa quotidiana*. Volume I: *Gli economisti*. Milano: FrancoAngeli.
- Baffigi, Alberto. 2010. "Teoria economica e legislazione sociale nel testo delle *Lezioni*." In Alfredo Gigliobianco, ed., *Luigi Einaudi: libertà economica e giustizia sociale*. Bari: Laterza, pp. 48–89.
- Barié, Ottavio. 1972. Luigi Albertini. Torino: Utet.
- Becchio, Giandomenica. 2004. "Salvatore Cognetti de Martiis e il Laboratorio di Economia politica (1893–1901)." *Il Pensiero Economico Italiano* 12 (2): 11–23.
- Benadusi, Lorenzo, ed. 2011. *Il 'Corriere' nell'età liberale*. Volume II: Documenti, 1900–1925. Milano: RCS Libri.
- Berta, Giuseppe. 1995. "Introduzione." In *Dall'occupazione delle fabbriche al fascismo. La crisi italiana del primo dopoguerra nei verbali della Lega industriale di Torino. 1920–23*. Torino: Unione Industriale, pp. 11–68.
- ——. 2012. "Un archetipo del giornalismo economico." In Marzio A. Romani, ed., *Luigi Einaudi e il* "Corriere della Sera" 1894–1925. Milano: Fondazione Corriere della Sera, vol. I, pp. lxxxvii–cxix.

- Bianchi, Giulia, and Alberto Giordano. 2010. "Questione sociale, mercato e istituzioni: la collaborazione di Luigi Einaudi a *La Stampa*." *Il Pensiero Economico Italiano* 18 (1): 45–79.
- Bini, Piero. 2010. "Economia industriale e realtà di mercato nell'umanesimo liberale di Luigi Einaudi." In Alfredo Gigliobianco, ed., *Luigi Einaudi: libertà economica e giustizia sociale*. Bari: Laterza, pp. 105–137.
- Bobbio, Norberto. 1974. "Il pensiero politico di Luigi Einaudi." *Annali della Fondazione Luigi Einaudi* 8: 183–215.
- Bonelli, Franco. 1971. La crisi del 1907. Una tappa dello sviluppo industriale in Italia. Torino: Fondazione L. Einaudi
- Brizzi, Riccardo. 2012. "Mass media e politica: dal telegrafo a internet." In Stefano Cavazza and Paolo Pombeni, eds., *Introduzione alla storia contemporanea*. Bologna: Il Mulino, pp. 177–197.
- Bruins, Gijsbert W. J., Luigi Einaudi, Edwin E. A. Seligman, and Josiah Stamp. 1923. *Report on Double Taxation Submitted to the Financial Committee*. Geneva: League of Nations.
- Cafagna, Luciano. 2004. "Luigi Einaudi, 1874–1961." Moneta e Credito 57 (March): 77–89.
- Caffè, Federico. 1987. "Einaudi, Luigi." In The New Palgrave: A Dictionary of Economics. Volume II. London: Macmillan, pp. 123–124.
- Candeloro, Giorgio. 1981. Storia dell'Italia moderna. Volume VII: La crisi di fine secolo e l'età giolittiana 1896–1914. Milano: Feltrinelli.
- Cassata, Francesco, and Roberto Marchionatti. 2010. "Cronache economiche di un trentennio. Lo sviluppo dell'economia italiana 1881–1913 nell'interpretazione di Einaudi e la sua Scuola." *Rivista di Storia Economica* 26 (2): 161–207.
- Castronovo, Valerio. 1976. La stampa italiana dall'Unità al fascismo. Bari: Laterza.
- . 1987. La Stampa 1867–1925: un'idea di democrazia liberale. Milano: FrancoAngeli.
- ——. 2017. La Stampa. 150 anni (1867–2017). Un giornale, la sua epoca, il suo futuro. Torino: Aragno Editore.
- Castronovo, Valerio, and Nicola Tranfaglia. 1979. Storia della stampa italiana. Volume III: La stampa italiana nell'età liberale. Roma and Bari: Laterza.
- Ciocca, Pierluigi, and Adalberto Ulizzi. 1990. "I tassi di cambio nominali e 'reali' dell'Italia dall'Unità nazionale al sistema monetario europeo (1861–1979)." In *Ricerche per la storia della Banca d'Italia*. Volume I. Bari: Laterza, pp. 341–368.
- Colarizi, Simona. 2011. Il 'Corriere' nell'età liberale. Volume I: Profilo storico. Milano: RCS Libri.
- Da Empoli, Domenico, Corrado Malandrino, and Valerio Zanone. 2014. "Editors' Introduction." In Domenico Da Empoli, Corrado Malandrino, and Valerio Zanone, eds., *Luigi Einaudi: Selected Political Essays*. Volume III. Houndmills: Palgrave Macmillan, pp. 1–27.
- De Cecco, Marcello. 2004. "Einaudi commentatore economico dell'età giolittiana." Rivista di Storia Economica 20 (3): 257–270.
- Duggan, Christopher. 2008. *The Force of Destiny. A History of Italy since 1796*. London: Penguin Books. Edwards, Ruth D. 1993. *The Pursuit of Reason: The Economist 1843–1993*. London: Hamish Hamilton.
- Einaudi, Luca, Riccardo Faucci, and Roberto Marchionatti. 2006. "Editors' Introduction." In Luca Einaudi, Riccardo Faucci, and Roberto Marchionatti, eds., *Luigi Einaudi. Selected Economic Essays*. Basingstoke: Palgrave Macmillan, pp. 1–25.
- Einaudi, Luigi. 1900a. *Un principe mercante. Studio sull'espansione coloniale italiana*. Torino: Bocca. ———. 1900b. *La rendita mineraria*. Torino: Utet.
- . 1901. "Il riconoscimento delle leghe all'estero." La Stampa, July 27, p. 1.
- . 1902. Studi sugli effetti delle imposte. Contributo allo studio dei problemi tributari municipali. Torino: Bocca.
- ——. 1905a. "L'espansione commerciale italiana." Corriere della Sera, May 8, p. 1.
- . 1905b. "La speculazione automobilistica." Corriere della Sera, June 9, p. 1.
- ——. 1905c. "Per le nuove convenzioni marittime." Corriere della Sera, November 21, p. 1.
- ——. 1906a. "Oppositori meridionali." Corriere della Sera, March 21, p. 1.
- ——. 1906b. "Per le nuove convenzioni marittime. Il passato." Corriere della Sera, May 14, p. 1.

——. 1906c. "Le nuove convenzioni marittime. Le proposte della Commissione reale." Corriere della	a
Sera, May 15, p. 1.	
——. 1906d. "Le proposte del governo per i servizi marittimi. I problemi insoluti." Corriere della Sera	ı,
May 21, p. 1.	
— . 1906e. "Il momento di Borsa." <i>Corriere della Sera</i> , November 3, p. 1.	
——. 1906f. "L'odierna crisi di borsa ed i suoi insegnamenti." <i>Corriere della Sera</i> , November 20, p. 1	
——. 1906g. "Emigrazione crescente." <i>Corriere della Sera</i> , December 12, p. 1.	
———. 1907. "Il ciclone americano." <i>Corriere della Sera</i> , November 18, p. 1.	
. 1910. "Il risorgimento del Mezzogiorno. Un popolo che si redime emigrando." <i>Corriere della Sera</i>	,
October 2, p. 1.	ι,
——. 1911a. "I trivellatori di stato." <i>La Riforma Sociale</i> 18 (Jan.): 1–14.	
——. 1911b. "Nuovi favori ai siderurgici?" <i>La Riforma Sociale</i> 18 (Feb.): 97–112.	
. 1911c. "An Italian Explanation of the Raid on Tripoli." <i>The Economist</i> 73, 21 (Oct.): 883–884.	
	a
Sera, February 21, p. 4.	
. 1912b. "E' possibile frenare il rincaro della vita e il ribasso delle rendite pubbliche?" Corriere della	a
Sera, February 27, p. 4.	
——. 1914a. "L'industria zuccheriera e il regime della protezione." <i>Corriere della Sera</i> , May 26, p. 1.	
. 1914b. "Un buon investimento e un'opera patriottica." <i>Corriere della Sera</i> , December 20, p. 2.	
——. 1915a. "La sottoscrizione al prestito nazionale." Corriere della Sera, January 3, p. 4.	
——. 1915b. "Il prestito nazionale. La proroga della sottoscrizione. Intorno a consigli e proposte d	li
lettori." Corriere della Sera, July 10, p. 2.	
——. 1915c. "Gli ammaestramenti del prestito di guerra." Corriere della Sera, July 22, p. 2.	
1915d. "Il trattamento dei primi e degli attuali sottoscrittori al prestito nazionale. Desideri de	ŀ
pubblico e domande di schiarimenti al Governo." Corriere della Sera, December 30, p. 2.	
	5
(Dec.): 493–511.	
	ı,
April 9, p. 2.	
. 1916b. "Restrizione di consumi, imposte ed azione di governo." Corriere della Sera, Septembe	r
25, p. 2.	
— . 1917a. "Opera patriottica e vantaggio finanziario." <i>Corriere della Sera</i> , January 22, p. 1.	
———. 1918a. "I denari nelle calze di lana." <i>Corriere della Sera</i> , February 3, p. 2.	
———. 1918b. "Bisogna sottoscrivere ancora!" <i>Corriere della Sera</i> , February 28, p. 2.	
———. 1919a. "Abolire i vincoli!" <i>Corriere della Sera</i> , January 15, p. 1.	
——. 1919b. "La politica del Tesoro. Abolire l'Istituto dei cambi!" <i>Corriere della Sera</i> , March 23, p. 1	
	•
——. 1919c. Hadotalize proteziolistiche. Corriere della sera, November 12, p. 1. ——. 1920a. Prediche. Bari: Laterza.	
——. 1920a. Prediche. Ball. Latetza. ——. 1920b. "Come si giunse al trattato di Versailles (Dal libro di un economista)." Corriere della Sera	
	ι,
February 15, pp. 1–2.	
——. 1920c. "Liquidare." <i>Corriere della Sera</i> , March 11, p. 1.	
. 1920d. "La contesa metallurgica. Principi fondamentali." <i>Corriere della Sera</i> , September 2, p. 1.	
——. 1921. Gli ideali di un economista. Firenze: La Voce.	
——. 1922a. "Il nuovo libro di Keynes." <i>Corriere della Sera</i> , January 26, p. 1.	
——. 1922b. "I risultati finanziari di Genova [comment to Keynes, 1922]." Corriere della Sera, Apri	il
29, pp. 1–2.	
——. 1922c. "Riabbeverarsi alla sorgente." <i>Corriere della Sera</i> , September 6, p. 1.	
——. 1922d. "The Financial Prospects of Italy." <i>Manchester Guardian Commercial</i> 1 (5): 285–288.	
——. 1923a. "La rivalutazione della lira." <i>Corriere della Sera</i> , June 20, p. 1.	
——. 1923b. "Per l'ultimo sforzo (in vista del pareggio)." Corriere della Sera, August 12, p. 1.	
. 1924. "Il silenzio degli industriali." Corriere della Sera, August 6, p. 1.	

- -. 1928. "Dei concetti di liberismo economico e di borghesia e sulle origini materialistiche della guerra." La Riforma Sociale 35 (Sept.-Oct.): 501-16. —. 1933a. "Il mio piano non è quello di Keynes." La Riforma Sociale 44 (March-April): 129-142. —. 1933b. Saggi. Torino: La Riforma Sociale. —. 1933c. La condotta economica e gli effetti sociali della guerra italiana. Bari: Laterza; New Haven: Yale University Press. —. 1941. "Intorno al contenuto dei concetti di liberismo, comunismo, interventismo e simili." Argomenti 9 (Dec.): 18-24. —. 1945a. "The Future of the Italian Press." Foreign Affairs 23 (3): 505–509. ——. 1945b. "Il problema dei giornali." Nuova Antologia 434 (July): 193–216. Einaudi, Luigi. [1949] 1964. Lezioni di politica sociale. Torino: Giulio Einaudi. ———. 1954. Il buongoverno. Saggi di economia e politica (1897–1954). Bari: Laterza. —. 1959. Cronache economiche e politiche di un trentennio. Volume I: 1893-1902. Torino: Giulio Einaudi. —. 1960. Cronache economiche e politiche di un trentennio. Volume III: 1910–1914. Torino: Giulio Einaudi. —. 1961. Cronache economiche e politiche di un trentennio. Volume V: 1919–1925. Torino: Giulio Einaudi. —. 1962. "Politicians and Economists." *Il Politico* 27 (2): 253–263. Faucci, Riccardo. 1986. Luigi Einaudi. Torino: Utet. –. 1993. "Einaudi, Luigi." In Dizionario biografico degli italiani. Volume XLII. Roma: Istituto della Enciclopedia Italiana, pp. 363-377. —. 1995. "Economia, storia, positivismo. Cognetti de Martiis e le origini del Laboratorio di economia

politica di Torino." Società e Storia 69: 599-618.

Faucci, Riccardo, and Giandomenica Becchio. 2010. "Einaudi e Albertini giornalisti." *Il Pensiero Economico Italiano* 18 (1): 103–124.

—. 2004. "La Scuola di Torino e il pensiero economico italiano." Il Pensiero Economico Italiano 12 (2):

- Fausto, Domenico. 1993. "La politica fiscale dalla prima guerra mondiale al regime fascista." In Franco Cotula, ed., *Ricerche per la storia della Banca d'Italia*. Volume II: *Problemi di finanza pubblica tra le due guerre*, 1919–1939. Bari: Laterza, pp. 3–138.
- Forgacs, David. 1990. *Italian Culture in the Industrial Era.* 1880–1980. Manchester: Manchester University Press.
- Forte, Francesco. 1982. Luigi Einaudi: il mercato e il buongoverno. Torino: Einaudi.
- . 2009. L'economia liberale di Luigi Einaudi. Saggi. Firenze: Olschki.
- Forte, Francesco, and Roberto Marchionatti. 2012. "Luigi Einaudi's Economics of Liberalism." *European Journal for the History of Economic Thought* 19 (4): 587–624.
- Frassati, Luciana. 1978–82. *Un uomo, un giornale. Alfredo Frassati*. Four volumes. Roma: Ed. di Storia e Letteratura.
- Fratianni, Michele, and Franco Spinelli. 1997. A Monetary History of Italy. Cambridge University Press.
- Giordano, Alberto. 2006. Il pensiero politico di Luigi Einaudi. Genova: Name edizioni.
- Gramsci, Antonio. 1975. *Quaderni del carcere*. Edited by Valerio Gerratana. Volume III: *Quaderni 12–19*. Torino: Einaudi.
- Heritier, Paolo, and Paolo Silvestri, eds. 2012. *Good Government, Governance, Human Complexity. Luigi Einaudi's Legacy and Contemporary Societies*. Firenze: Olschki.
- Keynes, John Maynard. 1922. "I risultati finanziari di Genova." Corriere della Sera, April 29, p. 1.
- Leoni, Bruno. 1964. "Luigi Einaudi e la scienza del governo." Il Politico 29 (1): 69-86.
- Mack Smith, Denis. 1997. Modern Italy: A Political History. Ann Arbor: The University of Michigan Press.

- Marchionatti, Roberto, ed. 2000. From Our Italian Correspondent: Luigi Einaudi's Articles in The Economist, 1908–1946. Two volumes. Firenze: Olschki.
- . 2017. "On the Relationship between Einaudi and Keynes in the Early 1920s." *Annals of the Fondazione Luigi Einaudi* 51 (Dec.): 267–286.
- Marchionatti, Roberto, Roberto Cassata, Giandomenica Becchio, and Fiorenzo Mornati. 2013. "When Italian Economics 'Was Second to None'. Luigi Einaudi and the Turin School of Economics." *European Journal of the History of Economic Thought* 20 (5): 776–811.
- Masini, Fabio. 2012. "Luigi Einaudi and the Making of the Neoliberal Project." *History of Economic Thought and Policy* 1 (1): 39–59.
- Mattei, Clara Elisabetta, 2017. "Austerity and Repressive Politics: Italian Economists in the Early Years of the Fascist Government." *European Journal of the History of Economic Thought* 24 (5): 998–1026.
- Monti, Mario. 1990. "Introduzione." In L. Einaudi, *Il mestiere della moneta*. Torino: Utet, pp. xv-xxviii.
- Moroni, Andrea. 2005. Alle origini del "Corriere della Sera". Da Eugenio Torelli Viollier a Luigi Albertini (1876–1900). Milano: FrancoAngeli.
- Murialdi, Paolo. 1997. Storia del giornalismo italiano. Bologna: Il Mulino.
- Nerozzi, Sebastiano. 2021. Luigi Einaudi scrittore di banca e di borsa. Un saggio introduttivo. Roma: Bancaria editrice.
- Parisi, Daniela. 2010. "Giovanni Demaria and the Rockefeller Foundation: Seesaw Relationship during a Thirty-Year Span." *Storia del Pensiero Economico* 7 (2): 81–91.
- Parsons, Wayne. 1989. The Power of the Financial Press. Journalism and Economic Opinion in Britain and America. Aldershot: Elgar.
- Pavanelli, Giovanni. 2006. "The Early Reception and Diffusion of Irving Fisher's Work in Italy." *Journal of the History of Economic Thought* 28 (3): 267–294.
- 2012. "Formare l'Italia nuova: gli articoli di Einaudi sul *Corriere della Sera* (1903–1925)." In Marzio A. Romani, ed., *Luigi Einaudi e il "Corriere della Sera*" 1894–1925. Two volumes. Volume I. Milano: Fondazione Corriere della Sera, pp. cxxi–clxxiii.
- 2016. "L'attività giornalistica di Luigi Einaudi in età liberale (1896–1925)." In Massimo M. Augello, Marco E. L. Guidi, and Giovanni Pavanelli, eds., Economia e opinione pubblica nell'Italia liberale. Gli economisti e la stampa quotidiana. Volume I: Gli economisti. Milano: FrancoAngeli, pp. 183–209.
- Pavanelli, Giovanni, and Giulia Bianchi. 2020. "The Italian Economists as Legislators and Policymakers During the Fascist Regime." In Massimo M. Augello, Marco E. L. Guidi, and Fabrizio Bientinesi, eds, An Institutional History of Italian Economics in the Interwar Period. Volume II: The Economics Profession and Fascist Institutions. Cham: Palgrave Macmillan, pp. 143–177.
- Pavanelli, Giovanni, and Chikako Nakayama. 2008. "A Lifelong Friendship: The Correspondence between Oskar Morgenstern and Luigi Einaudi." *Storia del Pensiero Economico* 5 (1): 95–120.
- Porta, Pier Luigi. 2010. "Libertà, mercato, giustizia sociale." In Alfredo Gigliobianco, ed., *Luigi Einaudi: libertà economica e giustizia sociale.* Bari: Laterza, pp. 19–47.
- Ricci, Umberto. 1917. "Sulla opportunità di una storia dell'economia politica italiana." In *In onore di Tullio Martello. Scritti varii*. Bari: Laterza, pp. 377–394.
- Romani, Marzio A. 2012a. "Vite parallele: Luigi Einaudi–Luigi Albertini." In Marzio A. Romani, ed., *Luigi Einaudi e il "Corriere della Sera*" 1894–1925. Two volumes. Volume I. Milano: Fondazione Corriere della Sera, pp. xi–lxxxvi.
- ———, ed. 2012b. *Luigi Einaudi e il "Corriere della Sera*" 1894—1925. Two volumes. Milano: Fondazione Corriere della Sera.
- Romeo, Rosario. 1974. "Luigi Einaudi e la storia delle dottrine e dei fatti economici." *Annali della Fondazione Luigi Einaudi* 8: 121–180.
- Salvadori, Massimo L. 2005. "Einaudi e la sua concezione del conflitto sociale." In *Luigi Einaudi: Istituzioni, mercato e riforma sociale.* Atti dei Convegni Lincei 214. Roma: Bardi, pp. 29–53.
- Silvestri, Paolo. 2008. Il liberalismo di Luigi Einaudi o del Buongoverno. Soveria Mannelli: Rubbettino.

- Skidelsky, Robert. 1992. *John Maynard Keynes*. Volume II: *The Economist as Saviour. 1920–1937*. New York: Penguin Books.
- Tedesco, Luca. 2021. La proposta antiprotezionista. I liberisti in Italia dalla crisi di fine Ottocento al fascismo. Milano: Mondadori.
- Tusset, Gianfranco. 2016. "Genova 1907: una crisi di borsa." In Massimo M. Augello, Marco E. L. Guidi, and Giovanni Pavanelli, eds., *Economia e opinione pubblica nell'Italia liberale. Gli economisti e la stampa quotidiana*. Volume II: *I dibattiti*. Milano: FrancoAngeli, pp. 179–201.
- Vivarelli, Roberto. 1981. *Il fallimento del liberalismo. Studi sulle origini del fascismo.* Bologna: Il Mulino. Zamagni, Vera. 1993. *The Economic History of Italy: 1860–1990.* Oxford: Clarendon.